Author’s response to reviews

Title: Evaluation of the thyroid characteristics of patients with growth hormone-secreting adenomas

Authors:

Dianshuang Xu (dickyhui@126.com)
Bolin Wu (497422449@qq.com)
Xiaoju Li (johnsylee@foxmail.com)
Yunju Cheng (775352197@qq.com)
Dubo Chen (chdub@mail3.sysu.edu.cn)
Yuefeng Fang (alex.fang@gdufs.edu.cn)
Qiu Du (duqiu@mail.sysu.edu.cn)
Zhiyong Chen (chenzhy43@outlook.com)
Xiaodong Wang (13922795840@163.com)

Version: 3  Date: 22 Aug 2019

Author’s response to reviews:

1. Funding.

-In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We have clarified the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript in the “Funding” section.

The study was financially supported by grants from the National Natural Scientific Foundation of China (No. 81600260) and the Guangdong Natural Science Foundation (No. 2016A030313210). National Natural Scientific Foundation of China (No. 81600260) was used to collect, analyze, and interpret of data. Guangdong Natural Science Foundation (No. 2016A030313210) was used to analyzed data and write manuscript.
2. Clean manuscript.

- The manuscript document contains a duplicate of the manuscript. Please remove this duplicate.

Response: We upload the clean revised manuscript only.